EO-1 Weekly Status Week of January 17 – January 23, 2008
Day of Year 017 - 023
Mission Day 2621 - 2627
Earth Observing-One (EO-1) General

There were 96 Data Collection Events (DCEs) scheduled this week.  
EO-1 INSTRUMENTS

· ALI internal calibration was performed on January 22 at 00:29z.

· The Advanced Land Imager (ALI) performed an outgassing from 01/19/2008 at 01:55z to 01/19/2008 at 16:45z.

· The Hyperion hyperspectral imager (HSI) performed a deicing from 01/19/2008 at 01:45z to 01/19/2008 at 16:35z.  

· New ALI flood lamp calibration sequence

· New sequence generated as an ATS load.

· Load has been verified and approved by the science group and the instrument engineers.

· FOT will utilize the Flight Software Lab  to test the ATS load prior to uplink.

· Execution of the sequence will be performed during a real-time GN contact to provide visible telemetry. 
· Two Lunar calibrations performed.
· Nominal Lunar calibration was performed during 02:45z-03:17z on January 22, 2008.

· Modified Lunar calibration was performed during 06:02z-06:35z on January 22, 2008.

· AC instrument was not a part of this calibration.

· Same settings as the November Lunar calibration were applied to see if the results would be consistent.

· Once the analyses of the results are completed by the science and technology group, the FOT will continue work with pointing and timing adjustments of the instrument commands.

EO-1 Spacecraft Subsystems

All spacecraft subsystems functioned nominally with the exception of the Command & Data Handling (C&DH) telemetry playback anomaly reported previously.  

Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations

A teleconference was held between the Advanced Information System Technology (AIST) Sensor Web collaborators on Friday, January 18.  The following items of interest were addressed:
· Need to investigate the implementation of a unified API (WFXML) that can execute any one of three workflow engines (i.e. BPEL, SensorML based, & OpenWFE).
· Work being conducted by Dryden Flight Research Center and West Virginia High Technology Consortium to control the Ikhana UAS autonomously.

· Need to follow up on Ames work to produce a SPS sensor service to control the UAS.

· Need for status report on SensorML modeling work.

· Need for cloud cover service (SPS) from Draper Lab to support the upcoming OGC Web Services 5 (OWS-5) testbed in March 2008.
· Need to identify contact at the National Weather Service that can supply RAWS interface information.

· Need to identify availability of EO-1 to support Rhodamine dye detection experiment the end of January 2008.

· Need to meet with the new GSFC IT security person to discuss what is required for OpenID to satisfy NASA’s security specifications.

Autonomous Sciencecraft Experiment (ASE)

With the exception of the periods for execution of the lunar calibrations and inclination burn, the ASE controlled EO-1 all week.

EO-1 MISSION OPERATIONS CENTER
Real-Time
· Inclination burn executed to increase MLT on January 17, 2008.

· A TDRSS contact was scheduled between 13:25z and 14:15z to provide visible telemetry to the FOT.

· See Flight Dynamics section for details on the maneuver.

Mission Planning
No change from last report.
Flight Dynamics
· Inclination burn executed.

· FOT designed and executed a 500 sec. inclination burn at 13:47:41z on January 17, 2008.

· Maneuver was executed on the ascending node of the orbit.

· Average duty cycle for the thrusters was 61.084%.

· Mean Local Time (MLT) is expected to rise above 10 am by mid April 2008.  

· The effect of the burns performed to date is expected to help maintain MLT above 10 am until approximately September 2008.

· After September 2008, the FOT will periodically schedule inclination burns which are smaller in magnitude to maintain MLT within desired times. 

· Flight Dynamics JRNOAA file update.

· FOT has failed to receive an updated JRNOAA file from the Flight Dynamics Facility (FDF) file server since January 19, 2008.

· FDF group has been notified of this issue.

Data Analysis and Trending
No change from last report.
System Administration
· Meeting was held on January 18, 2008 to discuss future work to be performed within the flight software lab.

· Honeywell will be responsible for all IT related work to be performed in the lab.

· The Flight Software Lab will be utilized to support end-of-life work.

· All code to support end-of-life activities will be tested and verified using the Flight Software Lab and virtual sat configurations prior to uplink.

· FOT will be involved in testing and verification of all flight code changes.

· Meeting was held on January 15 2008 to discuss the integration of ASPEN into the EO-1 control center.

· Project will order two new machines to support the efforts.

· One machine will be the web server for the ASPEN configuration.

· One machine will serve as the primary workstation.

· System administrator and the FOT will utilize another 32-bit machine as the complier.

· Supported developer efforts to revive MOPSS on the Axle server.

· VMware setup on Axle is currently incomplete.  

· Resolved Oracle issues on the Target server. 

Operational Enhancements Requests and Discrepancy Reports
No change from last report.
GROUND AND SPACE NETWORK

Station Downtime
· DataLynx Ground Station (PF1) X-Band is RED until further notice.

Operational Discrepancies
· PF1 experienced some problems on January 15, 2008 at both the X-Band downlinks at 06:12z and 15:19z which resulted in the loss of the following images:

· 2008:015:00:25:33-EO10980852008015110P0 Kangaroo Island [MSO/N]

· 2008:015:01:46:42-EO11140422008015110P1 Japan-Ryukyu Islands 2 [MDLS/N]

· 2008:015:03:31:03-EO11310562008015110K1 Indonesia-Sumatra [MDLS/E]

· 2008:015:05:09:25-EO11460552008015110P1 Maldives 23 [MDLS/N]

· 2008:015:18:09:02-EO10400332008015110P0 Railroad Valley 2005 [Ong/N]

Support Enhancements  
· Continuing to provide contacts to the Solar Dynamics Observatory Ground Systems (SDOGS) antennas.
· The SDO FOT will put together a series of tests to verify command interfaces between the SDO ground systems and the EO-1 spacecraft.
· Bit-wise comparison of the command ACRESPONDNOOP is complete.
· A write up will be presented to both projects including the details of the work performed to validate the ground systems.
· FOT will schedule a contact to support the command test.
· Contact tentatively scheduled for January 30, 2008 at 16:14z.
Upcoming Events
· New ALI flood lamp calibration to be performed in near future.

Imagery Status
Scenes and Engineering Cals planned for week of Jan 17 – Jan 23, 2008              96
Total scenes and engineering calibrations planned for entire mission – approximately 48,056
Total Scenes:  ALI scenes in the Level 0 archive            34,449 (as of Jan 23, 2008)

                        Hyperion scenes in the Level 0 archive    34,197
Publications and Presentations Status (as of 12-05-07)
404 publications 

284 external presentations

53 articles and press releases
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